Author's response to reviews

Title: Acute hyponatremia after cardioplegia by histidine tryptophane ketoglutarate a retrospective study

Authors:

Gregor Lindner (lindner.gregor@gmail.com)
Bernhard Zapletal (bernhard.zapletal@meduniwien.ac.at)
Christoph Schwarz (christoph.schwarz@medunigraz.at)
Wilfried Wisser (wilfried.wisser@meduniwien.ac.at)
Michael Hiesmayr (michael.hiesmayr@meduniwien.ac.at)
Andrea Lassnigg (andrea.lassnigg@meduniwien.ac.at)

Version: 2 Date: 28 March 2012

Author's response to reviews: see over
Bern, March 28th 2012

Type of article: Original investigation

Dear Prof. Zamvar,

Dear Prof. Taggart,

Dear Editors,

we are happy to submit the revised version of our manuscript “Acute hyponatremia after cardioplegia by histidine-tryptophane-ketoglutarate – a retrospective study” to the Journal of Cardiothoracic Surgery.

We are glad to let you know that we were able to address all points proposed by the reviewers.

We look forward to hear from you!

Sincerely,

Gregor Lindner for the authors